Author’s response to reviews

Title: case report: 1-year follow-up of cerebral sparganosis mansoni with a stroke-like onset

Authors:

DAN XIE (danxiedx@126.com)
Min WANG (dami_001@126.com)
XU CHEN (chenxu605@163.com)
ZHEN HOU TUO (smart1226@sina.com)

Version: 1 Date: 04 May 2018

Author’s response to reviews:

1. Abbreviations √

2. Ethics approval and consent to participate

The patient have been informed and signed the consent form.

3. Consent for publication

Yes, all the authors have signed for the publication.

4. Availability of data and material

All the data and material were collected by clinical.

5. Competing interests

Not applicable
6. Funding NO

7. Authors’ contributions

Dr. Dan XIE collected the patient’s materials and did the writings; Dr Min WANG was the competent doctor during the hospitalization; Dr Xu CHEN was assisted with the diagnosis and follow-up; Dr Hou-zhen TUO was the superior physician who guided diagnosis and treatment.

8. Acknowledgements

The authors would like to thank the department of neurosurgery, the department of imaging, and the tropical disease research institute for their help and contribution in the diagnosis, the treatment, and the follow-up of the patient in this study.